Author's response to reviews

Title: Recommendations to improve physical activity among teenagers- A qualitative study with ethnic minority and European teenagers

Authors:

Sinead Brophy (s.brophy@swansea.ac.uk)
Annie Crowley (a.crowley@africaeducationaltrust.org)
Rupal Mistry (R.Mistry@comicrelief.com)
Rebecca Hill (r.a.hill@swanse.ac.uk)
Sopna Choudhury (s.m.choudhury@bham.ac.uk)
Non E Thomas (n.e.thomas@swanse.ac.uk)
Frances Rapport (f.l.rapport@swanse.ac.uk)

Version: 2 Date: 11 January 2011

Author's response to reviews:

Dear Editor,

Thank you for considering this manuscript.

The editor of BMC Public Health asked "Prior to submitting to BMC Public Health we request make clearer the distinction between nationality and ethnicity in your manuscript".

We have made the following changes:
We had added in the results section the number of participants born in the UK (i.e. UK national but ethnicity Somali/Bangladeshi/European etc).

We have added in the discussion a note about being born in the UK "...be a migration factor, in that the Bangladeshi children were generally born in the UK"

with very good wishes and thanks

Dr Sinead Brophy